Author's response to reviews

Title: Community pharmacists' involvement in smoking cessation: implementation of the National smoking cessation guideline in Finland

Authors:

Terhi Kurko (terhi.a.kurko@helsinki.fi)
Kari Linden (kari.linden@pfizer.com)
Kirsi Pietilä (kirsi.pietila@apteekit.net)
Patrick Sandström (patrick.sandstrom@thl.fi)
Marja Airaksinen (marjaairaksinen@gmail.com)

Version: 6 Date: 29 July 2010

Author's response to reviews:

Dear Editors:

With this final proof stage of the manuscript, I only changed couple of spelling mistakes, (=correction of the typogrpahy). I still noticed in the main text. I made some corrections in the Tables 1 and 4 (pages 26-27 and 31). In Table 1 I divided the text so, that the table continues in the page 27 from a new heading.

Some words were missing in the Table 4 (page 31), because the Table had been altered with narrower rows and this change took away the words we had in the earlier versions of this table. I added now the missing words.

None of the changes made altered the content of the manuscript.

With best regards: Terhi Kurko